Author's response to reviews

Title: BMI and dental caries in children and adolescents: a systematic review of the literature

Authors:

Merrilyn Hooley (merrilyn@deakin.edu.au)
Helen Skouteris (helen.skouteris@deakin.edu.au)
Cecile Boganin (cecile.boganin@deakin.edu.au)
Julie Satur (juliegs@unimelb.edu.au)
Nicky Kilpatrick (nicky@bassdata.com.au)

Version: 2 Date: 23 June 2012

Author's response to reviews: see over
June 23rd, 2012

Drs Moher, Shekelle, and Stewart, Editors in Chief
Systematic Reviews
c/o BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Re: MS: 213486655663594 BMI and dental caries in children and adolescents: a systematic review of the literature

Dear Drs Moher, Shekelle, and Stewart,

We appreciate the opportunity to resubmit our manuscript “Obesity and dental caries in children and adolescents: a systematic review of the literature” to your journal Systematic Reviews.

We are grateful to the reviewers for their very thoughtful suggestions on our earlier draft and have addressed each of the concerns. We believe our manuscript is much improved as a result of this process and would like to extend our appreciation to you and to the reviewers. With their permission, we would like to acknowledge their valuable feedback on our earlier draft and have included this in our current manuscript.

We hope you find our review improved and of interest to your readers, and look forward to your response. Please do not hesitate to contact me if I can provide any further information.

With kindest regards,

Merrilyn Hooley